\documentclass[12pt]{letter} \textwidth 7in\oddsidemargin -0.25in
\textheight 9.2in\topmargin -0.85in
\renewcommand{\arraystretch}{0.98} \thispagestyle{empty}
\renewcommand{\baselinestretch}{1.1}
\usepackage{amssymb}
\usepackage{amsmath}
\newtheorem{example}{Example}
\newtheorem{lem}{Lemma}
\newtheorem{co}{Corollary}
\newtheorem{remark}{Remark}
\newtheorem{pr}{Proposition}
\newtheorem{no}{Notation}
\newtheorem{con}{Conjecture}
\newenvironment{proof}{\noindent {\em Proof:}}{\hspace*{1cm}
        \hspace*{\fill}$\rule{1.2ex}{1.4ex}$\medskip}
\newenvironment{re}{\begin{remark}\rm}{\end{remark}}
\newenvironment{ex}{\begin{example}\rm}{\end{example}}
\newenvironment{de}{\begin{definition}\rm}{\end{definition}}
\newcommand{\un}[1]{\underline{{\em #1}}}
\newcommand{\ebox}{\framebox[4mm]{\rule[0mm]{0mm}{2mm}}}
\newcommand{\xbox}{\framebox[4mm]{\rule[0mm]{0mm}{2mm}x}}
%\newcommand{\ebox}{\fbox{\rule[-1mm]{0mm}{4mm}}}


\begin{document}
\begin{center}
 \large{Math 110, Spring 1999}
\end{center}

\bigskip

\textbf{Instructors:}  Gerard Misiolek and Bruce Williams

\textbf{Office:}  CCMB 362 (Misiolek) and CCMB 236 (Williams)

\textbf{Text:}  Chapters 9-11 and parts of chapter 14 of Hahn,
\textit{Basic Calculus:  From Archimedes to Newton to its Role in
Science.}

\bigskip

\textbf{Examinations:}
\begin{quote}
Quizzes:  There will be one quiz every two weeks, usually
consisting of two problems.

Midterm:  Wednesday, March 3, (in class)

Final:  Wednesday, May 6, 1:45 - 3:45 (place to be announced)

Note:  The University Honor Code applies to all the above
\end{quote}
\bigskip

\textbf{Homework:}  There will be a problem set due every
Friday with the final write up to be turned in on the following
Monday.  Some of the most difficult problems will be discussed
on Friday and I will ask for volunteers to present solutions on
the board.  Such presentations will receive extra credit
(depending on their quality).

\bigskip

\textbf{Grades} (Roughly):  Final Exam 30$\%$, Midterm 20$\%$,
Quizzes 30$\%$, Homework 20$\%$ 

\bigskip

\textbf{Topics:}
\begin{quotation}
From Engineering:  Basic facts about differentiation, a problem
from de L'Hospital's Calculus text, basic facts about forces and
tensions, the suspension bridge.\\

From Science:  Basics about the exponential and logarithm
functions, nuclear clocks and what they tell us, growth in
biology.\\

From Engineering:  Review of integration and graphing, work and
energy, interior ballistics, the Springfield rifle.\\
\end{quotation}

\end{document}